Author's response to reviews

Title: Intrahepatic Biliary Mucinous Cystic Neoplasms: Clinicoradiological Characteristics and Surgical Results

Authors:

Chao-Wei Lee (alanchaoweilee@hotmail.com)
Hsin-I Tsai (tsaic@hotmail.com)
Yann-Sheng Lin (yann.mail@msa.hinet.net)
Tsung-Han Wu (wutsunghan@gmail.com)
Ming-Chin Yu (mingchin2000@gmail.com)
Miin-Fu Chen (chenmf@adm.cgmh.org.tw)

Version: 2 Date: 14 November 2014

Author's response to reviews:

We have included 1. line and page numbering, 2. email addresses of all authors, 3. Authors' Contribution section, and 4. Competing Interest section in the main manuscript.